Author's response to reviews

Title: A Cluster-Randomized, Placebo-controlled, Maternal Vitamin A or Beta-Carotene Supplementation Trial in Bangladesh: Design and Methods

Authors:

Alain B Labrique (alabriqu@jhsph.edu)
Parul Christian (pchristi@jhsph.edu)
Rolf D.W. Klemm (rklemm@jhsph.edu)
Mahbubur Rashid (mrashid.jivita@gmail.com)
Abu Ahmed Shamim (shamim.jivita@gmail.com)
Allan Massie (amassie@jhsph.edu)
Kerry Schulze (kschulze@jhsph.edu)
Andre Hackman (ahackman@jhsph.edu)
Keith P West Jr (kwest@jhsph.edu)

Version: 4 Date: 7 April 2011

Author's response to reviews:

The abstracts in the system and the manuscripts have been made the same. Table references have been checked and corrected, where necessary.